\documentclass[10pt]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}

\def\udcs{510.67} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{03C07, 03C52} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem{fact}{Fact}
\newtheorem{note}{Remark}

\def\logo{{\bf\huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}

\def\semrtop
     {
  \vbox{
     \noindent\logo
     \hspace{80mm}\raisebox{1ex}{ISSN 1813-3304 }

     \vspace{5mm}

     \begin{center}
     {\huge СИБИРСКИЕ \ ЭЛЕКТРОННЫЕ} \\[2mm]  %Headings in Russian
     {\huge МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ} \\[2mm]
     {\large Siberian Electronic Mathematical Reports} \\[1mm]
     {\LARGE\tt{http://semr.math.nsc.ru}}\\[0.5mm]
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
     \vspace{-3mm}
     \noindent
     \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa} \\
     \hline\hline
     \end{tabular}

     \vspace{1mm}
     {\flushleft\it Том 19, стр. ???--??? (2022) \hspace{64mm}{\rm\small УДК \udcs}} %Volume, pages in Russian,
                                                                                    %filled by Editorial board
     \newline
         {\rm\small DOI~10.33048/semi.2019.16.xxx}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaa}{\rm\small MSC\ \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{99}

\bibitem{cs} S.V.~Sudoplatov, {\it Combinations of structures}, The Bulletin of Irkutsk State
University. Series Mathematics, {\bf 24} (2018), 65--84.

\bibitem{ccct} S.V.~Sudoplatov, {\it Combinations related to classes of finite
and countably categorical structures and their theories}, Siberian
Electronic Mathematical Reports, {\bf 14} (2017), 135--150.

\bibitem{cl} S.V.~Sudoplatov, {\it Closures and generating sets related to combinations of
structures}, The Bulletin of Irkutsk State University. Series
Mathematics, {\bf 16} (2016), 131--144.

\bibitem{lut} S.V.~Sudoplatov, {\it Families of language uniform theories and their generating sets},
The Bulletin of Irkutsk State University. Series Mathematics, {\bf
17} (2016), 62--76.

\bibitem{at} S.V.~Sudoplatov, {\it Approximations of
theories}, Siberian Electronic Mathematical Reports, {\bf 17}
(2020), 715--725.

\bibitem{AKS} A.B.~Altayeva, B.Sh.~Kulpeshov, S.V.~Sudoplatov, {\it $E$-combinations
of $\aleph_0$-categorical linear orders}, Journal of Mathematics,
Mechanics and Computer Science, {\bf 103}:3 (2019), 3--12.

\bibitem{KS_P1} B.Sh.~Kulpeshov, S.V.~Sudoplatov, {\it $P$-combinations of ordered theories},
Lobachevskii Journal of Mathematics, {\bf 41}:2 (2020), 227--237.

\bibitem{KS_P2} B.Sh.~Kulpeshov, S.V.~Sudoplatov, {\it $P^\ast$-combinations of almost
$\omega$-categorical weakly o-minimal theories}, Lobachevskii
Journal of Mathematics, {\bf 42}:4 (2021), 743--750.

\bibitem{KS_E} B.Sh.~Kulpeshov, S.V.~Sudoplatov, {\it $E$-combinations of almost omega-categorical
weakly $o$-minimal theories}, Algebra and model theory 13,
Collection of papers edited by A.G. Pinus, E.N. Poroshenko, and
S.V. Sudoplatov, Novosibirsk State Technical University,
Novosibirsk, 2021.~--- P.~83--89.

\bibitem{Rosen} E.~Rosen, {\it Some Aspects of Model Theory and Finite
Structures}, The Bulletin of Symbolic Logic, {\bf 8}:3 (2002),
380--403.

\bibitem{Vaa} J.~V\"{a}\"{a}n\"{a}nen, {\it Pseudo-finite model theory}, Matematica Contemporanea,
{\bf 24} (2003), 169--183.

\bibitem{ChHr} G.~Cherlin, E.~Hrushovski, {\it Finite Structures with Few Types}, Annals of
Mathematics Studies, No. 152, Princeton University Press,
Princeton, Oxford, 2003.

\bibitem{MS} H.D.~Macpherson, Ch.~Steinhorn, {\it Definability in the classes of finite
structures}, Finite and Algorithmic Model Theory. London
Matheematical Society Lecture Notes series: 379~/ eds.:
J.~Esparza, C.~Michaux, Ch.~Steinhorn, Cambridge University Press,
Cambridge, 2011.~--- P.~140--176.

\bibitem{dst} N.D.~Markhabatov, S.V.~Sudoplatov, {\it Definable subfamilies of theories,
related calculi and ranks}, Siberian Electronic Mathematical
Reports, {\bf 17} (2020), 700--714.

\bibitem{MS19} N.D.~Markhabatov, S.V.~Sudoplatov, {\it Algebras for definable families of theories},
Siberian Electronic Mathematical Reports, {\bf 16} (2019),
600--608.

\bibitem{rest} S.V.~Sudoplatov, {\it Relative $e$-spectra and
relative closures for families of theories}, Siberian Electronic
Mathematical Reports, {\bf 14} (2017), 296--307.

\bibitem{RSrank} S.V.~Sudoplatov, {\it Ranks for families of theories
and their spectra}, Lobachevskii Journal of Mathematics, {\bf
42}:12 (2021), 2959--2968.

\bibitem{MarkhSud} N.D.~Markhabatov, S.V.~Sudoplatov, {\it Ranks for families of all theories
of given languages}, Eurasian Mathematical Journal, {\bf 12}:2
(2021), 52--58.

\bibitem{FPM} S.V.~Sudoplatov, {\it Formulas and properties, their links and
characteristics}, Mathematics, {\bf 9}:12 (2021), 16~pp.

\bibitem{PS21} In.I.~Pavlyuk, S.V.~Sudoplatov, {\it Formulas and properties for families
of theories of Abelian groups}, Bulletin of Irkutsk State
University. Series Mathematics, {\bf 36} (2021), 95--109.

\bibitem{M19} N.D.~Markhabatov, {\it Ranks for families of permutation theories},
The Bulletin of Irkutsk State University. Series Mathematics, {\bf
28} (2019), 86--95.

\bibitem{PS19} In.I.~Pavlyuk, S.V.~Sudoplatov, {\it Ranks for families of theories of abelian groups},
The Bulletin of Irkutsk State University. Series Mathematics, {\bf
28} (2019), 96--113.

\bibitem{KS_rank2019} B.Sh. Kulpeshov, S.V. Sudoplatov, {\it Ranks and approximations for families
of ordered theories}, Algebra and model theory 12, Collection of
papers edited by A.G. Pinus, E.N. Poroshenko, and S.V. Sudoplatov,
Novosibirsk State Technical University, Novosibirsk, 2019.~---
P.~32--40.

\bibitem{KPS_2021} B.Sh. Kulpeshov, In.I.~Pavlyuk, S.V. Sudoplatov,
{\it On criterion of total transcendency for families of ordered
theories}, Abstracts of International conference ''Maltsev
Meetings'', Novosibirsk, 2021, P.~152.

\bibitem{BaLa} J.T.~Baldwin, A.H.~Lachlan, {\it On strongly minimal sets}, The Journal
of Symbolic Logic, {\bf 36}:1 (1971), 79--96.

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}


\thispagestyle{empty}

\title[Properties of ranks]{Properties of ranks for families of strongly minimal theories}
\author{B.Sh. Kulpeshov, S.V. Sudoplatov}%
\address{Beibut Shaiykovich Kulpeshov
\newline\hphantom{iii} Kazakh British Technical University
\newline\hphantom{iii} Tole bi street, 59
\newline\hphantom{iii} 050000, Almaty, Kazakhstan.
}%
\email{b.kulpeshov@kbtu.kz}%
\address{Novosibirsk State Technical University
\newline\hphantom{iii} K. Marx avenue, 20
\newline\hphantom{iii} 630073, Novosibirsk, Russia.}
\email{kulpeshov@corp.nstu.ru}%

% .
%, , , ; .

\address{Sergey Vladimirovich Sudoplatov
\newline\hphantom{iii} Novosibirsk State Technical University
\newline\hphantom{iii} K. Marx avenue, 20
\newline\hphantom{iii} 630073, Novosibirsk, Russia.
}%
\email{sudoplatov@corp.nstu.ru}%
\address{Sobolev Institute of Mathematics,
\newline\hphantom{iii} Academician Koptyug avenue, 4
\newline\hphantom{iii} 630090, Novosibirsk, Russia.
}%
\email{sudoplat@math.nsc.ru}

\thanks{\sc Kulpeshov, B.Sh., Sudoplatov, S.V.,
Properties of ranks for families of strongly minimal theories}
\thanks{\copyright \ 2022 Kulpeshov B.Sh., Sudoplatov S.V.}
\thanks{\rm The research is supported by Russian Scientific
Foundation, Project No.~22-21-00044.}
\thanks{\it Received  December, ??, 2021, published  ???, ??,  2022.}%


\semrtop \vspace{1cm} \maketitle {\small
\begin{quote}
\noindent{\sc Abstract.} We study rank properties for families of
strongly minimal theories. A criterion for $e$-total
transcendence of families of strongly minimal theories is
obtained in terms of the description of the language
symbols.\medskip

\noindent{\bf Keywords:} strongly minimal theory, sentence, family of
theories, rank for a fa\-mi\-ly of theories.
 \end{quote}
}


\section{Introduction}

In the study of elementary theories, an important role is played
by the description of their interrelationships and derived
objects, including main characteristics. Essen\-t\-i\-al
interrelationships between theories arise when considering their
various com\-bi\-na\-t\-ions \cite{cs, ccct}, closures and
generating sets for families of theories \cite{cl, lut},
appro\-x\-i\-ma\-t\-ions of theories \cite{at}. Along with
combinations of theories in general, various combinations of
ordered theories are considered, which allow describing their
esse\-n\-t\-i\-al structural properties \cite{AKS, KS_P1, KS_P2,
KS_E}. Among various approximations of theories, one of the
central places is occupied by approximations by theories of finite
structures, which define pseudofinite structures \cite{Rosen, Vaa,
ChHr, MS}. Important derived objects are algebras of definable
subfamilies \cite{dst, MS19}. Main characteristics related to the
families of theories are the $e$-spectra \cite{cs, rest}, the
ranks for the families of theories \cite{dst, RSrank, MarkhSud}
and for the formulas \cite{FPM, PS21}. Rank values allow one to
determine how complex certain types of families of theories are,
both in general case and in relation to natural classes of
theories, including classes of substitution theories \cite{M19},
theories of abelian groups \cite{PS19}, ordered theories
\cite{KS_rank2019, KPS_2021}.

The present article is concerned the notion of {\it strong
minimality} initially deeply investigated in \cite{BaLa}. We study
rank properties for families of strongly minimal theories. A
criterion for $e$-total transcendence of families of strongly
minimal theo\-ries is obtained in terms of the description of the
language symbols.

\section{Preliminary notions}

\begin{definition} {\rm \cite{BaLa} Let $T$ be a complete theory without finite models.
The theory $T$ is said to be {\it strongly minimal} if for any model
$\mathcal{M}\models T$, any formula $\phi(x,\bar y)$ and any tuple
$\bar a\in M$ either the set $\phi(\mathcal{M},\bar a)$ is finite
or the set $\neg\phi(\mathcal{M},\bar a)$ is finite.}
\end{definition}

\begin{definition}\rm \cite{RSrank} Let $\mathcal{T}$ be a family of complete theories in a fixed
language $\Sigma$, $\phi$ be an arbitrary $\Sigma$-sentence. Then
the set $\mathcal{T}_{\phi}:=\{T\in\mathcal{T}\mid T\models
\phi\}$ is said to be the {\it $\phi$-neighbourhood} of the family
$\mathcal{T}$, or the {\it $s$-definable} subfamily of
$\mathcal{T}$ defined by the sentence~$\phi$.

We define the {\em rank} ${\rm RS}$ for the family $\mathcal{T}$
as follows:

(1) ${\rm RS}(\mathcal{T})=-1$, if $\mathcal{T}=\emptyset$.

(2) ${\rm RS}(\mathcal{T})=0$, if  $\mathcal{T}$ is a finite
nonempty family.

(3) ${\rm RS}(\mathcal{T})\ge 1$, if $\mathcal{T}$ is infinite.

(4) ${\rm RS}(\mathcal{T})\ge \alpha+1$, if there are pairwise
inconsistent $\Sigma$-sentences $\phi_n, n\in\omega$, such that
${\rm RS}(\mathcal{T}_{\phi_n})\ge \alpha$.

(5) If $\delta$ is a limit ordinal, then ${\rm RS}(\mathcal{T})\ge
\delta$, if  ${\rm RS}(\mathcal{T})\ge \beta$ for any
$\beta<\delta$.

We put ${\rm RS}(\mathcal{T})=\alpha$, if ${\rm
RS}(\mathcal{T})\ge\alpha$ and $\neg[{\rm
RS}(\mathcal{T})\ge\alpha+1]$.

If ${\rm RS}(\mathcal{T})\ge\alpha$ for any $\alpha$, then we put
${\rm RS}(\mathcal{T})=\infty$.

A family $\mathcal{T}$ is called {\em $e$-totally transcendental},
or {\em totally transcendental}, if ${\rm RS}(\mathcal{T})$ is an
ordinal.

If a family $\mathcal{T}$ is $e$-totally transcendental, with
${\rm RS}(\mathcal{T})=\alpha\geq 0$, then the {\em degree} ${\rm
ds}(\mathcal{T})$ of $\mathcal{T}$ is the maximal number of
pairwise inconsistent sentences $\phi_i$ with ${\rm
RS}(\mathcal{T}_{\varphi_i})=\alpha$.
\end{definition}

By the definition, if ${\rm RS}(\mathcal{T})=\alpha\in{\rm Ord}$
then ${\rm ds}(\mathcal{T})\in\omega\setminus\{0\}$.

\section{Families of strongly minimal theories and their ranks}

Throughout this article we denote by $\mathcal{T}_{\Sigma}$ the family of all
strongly minimal theories in a language $\Sigma$.

\begin{proposition}\label{prop1} Let $\Sigma^1_{\kappa}:=\{P^1_i\}_{i<\kappa}$, where
$\kappa$ is a cardinal. Then the following holds:

{\rm (1)} If $\kappa<\omega$ then ${\rm
RS}(\mathcal{T}_{\Sigma^1_{\kappa}})=\kappa$.

{\rm (2)} If $\kappa \ge \omega$ then ${\rm
RS}(\mathcal{T}_{\Sigma^1_{\kappa}})=\infty$.
\end{proposition}

\begin{proof}
(1) By virtue of strong minimality, each unary predicate selects a
finite or cofinite set in any model of an arbitrary strongly
minimal theory.

We construct an infinitely branching tree of length $\kappa$ for the
family $\mathcal{T}_{\Sigma^1_{\kappa}}$ of theories. At the first
level, we distinguish theories in which the first predicate $P_1$
selects in any model a finite set consisting of $n$ elements for
each $n <\omega$. Further, at the $i$-th level ($i\le\kappa$), we
distinguish theories in which the predicate $P_i$ selects a finite
set consisting of $n$ elements for each $n<\omega$. Therefore, we
obtain ${\rm RS}(\mathcal{T}_{\Sigma^1_{\kappa}})=\kappa$.

(2) Since there exist infinitely many unary predicates, then one
can construct an infinitely branching tree of infinite length,
whence ${\rm RS}(\mathcal{T}_{\Sigma^1_{\kappa}})=\infty$.
\end{proof}

Let $\Sigma^f_1:=\{f^1\}$, where $f$ is a unary functional symbol,
i.e. any theory $T$ of the language $\Sigma^f_1$ satisfies
the following sentence: $T\models (\forall x)(\exists! y) f(x)=y.$

Let $T$ be a complete theory of the language $\Sigma^f_1$,
$M\models T$.

We say that an element $a\in M$ {\it forms a cycle of length $1$}
if $f(a)=a$.

We say that elements $a_1, a_2, \ldots, a_n\in M$ {\it form a
cycle of length $n$} if $a_i\ne a_j$ for any $1\le i,j \le n$ with
$i\ne j$, $f(a_1)=a_2, f(a_2)=a_3, \ldots, f(a_{n-1})=a_n$ and
$f(a_n)=a_1$.

Consider the following sentences:
$$\theta^1_n:=\exists x_1 \ldots \exists x_n [\land_{i\ne j} x_i\ne x_j \land \land_{i=1}^{n-1} f(x_i)=x_{i+1} \land f(x_n)=x_1\land$$
$$\forall y_1 \ldots \forall y_n (\land_{i\ne j} y_i\ne y_j \land \land_{i=1}^{n-1} f(y_i)=y_{i+1}\land f(y_n)=y_1 \to \lor_{i=1}^n y_1=x_i)],$$
$$\theta^2_n:=\exists x_1 \ldots \exists x_n \exists y_1 \ldots \exists y_n[\land_{i\ne j} x_i\ne x_j \land \land_{i=1}^{n-1} f(x_i)=x_{i+1} \land f(x_n)=x_1\land$$
$$\land_{i\ne j} y_i\ne y_j \land \land_{i=1}^{n-1} f(y_i)=y_{i+1}\land f(y_n)=y_1\land$$
$$ \forall z_1 \ldots \forall z_n (\land_{i\ne j} z_i\ne z_j \land \land_{i=1}^{n-1} f(z_i)=z_{i+1}\land f(z_n)=z_1 \to \lor_{i=1}^n (z_1=x_i\lor z_1=y_i))].$$

The sentence $\theta^1_n$ asserts that there exists a unique cycle of
length $n$, and the sentence $\theta^2_n$ asserts that there exist
exactly two cycles of length $n$. Similarly, we can construct a
sentence $\theta^k_n$ asserting that there exist exactly $k$ cycles
of length $n$, for any $k,n<\omega$.

\begin{proposition}\label{prop2} Let $\Sigma^f_1:=\{f^1\}$, where $f$ is a unary functional symbol.
Then ${\rm RS}(\mathcal{T}_{\Sigma^f_1})=\infty$.
\end{proposition}

\begin{proof}
We construct an infinitely branching tree of infinite length for
$\mathcal{T}_{\Sigma^f_1}$.

At the first level, we distinguish theories in which there are
exactly $n$ cycles of length $1$ for each $n<\omega$. At the
second level, we distinguish theories in which there are exactly
$n$ cycles of length $2$ for each $n<\omega $. At the $k$-th level
($k<\omega$), we distinguish theories in which there are exactly
$n$ cycles of length $k$ for each $n <\omega $. Thus, we obtain
${\rm RS}(\mathcal{T}_{\Sigma^f_1})=\infty$.
\end{proof}

Since $k$-ary predicates and $k$-ary operations, for $k\geq 2$,
allow to interpret unary operations, Proposition \ref{prop2}
implies:

\begin{corollary}\label{cor01} If a language $\Sigma$ contains a $k$-ary predicate symbol or a $k$-ary fun\-c\-tio\-nal symbol, where $k\geq 2$,
then ${\rm RS}(\mathcal{T}_{\Sigma})=\infty$.
\end{corollary}

\begin{proposition}\label{propMS} {\rm \cite{MarkhSud}} If $\Sigma$ is a language consisting of constant symbols then for the family $\mathcal{T}$ of all theories
in the language $\Sigma$ either ${\rm RS}(\mathcal{T})=1$ and
${\rm ds}(\mathcal{T})=P(n)$, where $P(n)$ is the number of
partitions of $n$-element sets, if $\Sigma$ consists of
$n$ symbols, where $n<\omega$, or ${\rm RS}(\mathcal{T})=\infty$, if
$\Sigma$ has infinitely many symbols.
\end{proposition}

Proposition \ref{propMS} implies:

\begin{fact}\label{fact1} Let $\Sigma^c_{\kappa}:=\{c_i\}_{i<\kappa}$, where $\kappa$ is a cardinal.
Then ${\rm RS}(\mathcal{T}_{\Sigma^c_{\kappa}})=0$ and ${\rm
ds}(\mathcal{T}_{\Sigma^c_{\kappa}})=P(\kappa)$, where $P(\kappa)$
is the number of partitions of $n$-element set, if
$\kappa<\omega$, or ${\rm
RS}(\mathcal{T}_{\Sigma^c_{\kappa}})=\infty$, if
$\kappa\geq\omega$.
\end{fact}


\begin{proposition}\label{prop3}
Let $\Sigma^{1, c}_{k,m}:=\{P^1_1, \ldots, P^1_k, c_1, \ldots,
c_m\}$, where $k, m<\omega$. Then ${\rm
RS}(\mathcal{T}_{\Sigma^{1, c}_{k,m}})=k$ and ${\rm
ds}(\mathcal{T}_{\Sigma^{1, c}_{k,m}})=P'(k,m)$, where $P'(k,m)$
is the total number of actions by predicates $P^1_1, \ldots,
P^1_k$ on each of $P(m)$ partitions of the set
$\{c_1,\ldots,c_m\}$ on $r$ parts, $1\leq r\leq m$:
$P'(k,m)=\sum\limits_{r}2^{kr}.$
\end{proposition}

\begin{proof}
By virtue of Proposition \ref{prop1} and Fact \ref{fact1}, we
assert that ${\rm RS}(\mathcal{T}_{\Sigma^{1, c}_{k,m}})=k$. Let
us now understand that ${\rm ds}(\mathcal{T}_{\Sigma^{1,
c}_{k,m}})=2^{km}$. Each of the unary predicates $P^1_i$ can act
on the set of identified constants $c_j$ in two ways, i.e. either
$P^1_i(c_j)$, for all identified constants $c_j$, or $\neg
P^1_i(c_j)$ (each constant has only two possibilities for a chosen
unary predicate). Therefore, taking an arbitrary partition of the
set $\{c_1,\ldots,c_m\} $ into $r$ parts, we obtain the
multiplication $r$ times $2^k$ of the actions of the predicates
$P^1_1,\ldots,P^1_k$ on each element partition equal to $2^{kr}$.
Running through all the partitions of constants and all variants
of the influence of the predicates $P^1_1,\ldots,P^1_k $ on the
elements of the partitions of the set $\{c_1,\ldots,c_m\} $, we
obtain the desired equality ${\rm ds}(\mathcal{T}_{\Sigma^{1,
c}_{k,m}})=P'(k,m)$.
\end{proof}

\begin{proposition}\label{propMS2} {\rm \cite{MarkhSud}} If $\Sigma$
is a language of $0$-ary predicates, then for the family
$\mathcal{T}$ of all theories in the language $\Sigma$ either
${\rm RS}(\mathcal{T})=0$ and ${\rm ds}(\mathcal{T})=2^m$, if
$\Sigma$ consists of $m$ $0$-ary predicates, where $m<\omega$, or ${\rm
RS}(\mathcal{T})=\infty$, if $\Sigma$ has infinitely many symbols.
\end{proposition}

Proposition \ref{propMS2} implies:

\begin{fact}\label{fact2} Let $\Sigma^0_{\kappa}:=\{P^0_i\}_{i<\kappa}$,
where $\kappa$ is a cardinal. Then either ${\rm
RS}(\mathcal{T}_{\Sigma^0_{\kappa}})=0$ and ${\rm
ds}(\mathcal{T}_{\Sigma^c_{\kappa}})=2^\kappa$, if
$\kappa<\omega$, or ${\rm
RS}(\mathcal{T}_{\Sigma^0_{\kappa}})=\infty$, if
$\kappa\geq\omega$.
\end{fact}

In view of Fact \ref{fact2} if we add to the language $\Sigma$
finitely many $0$-ary predicate symbols then we preserve the
$e$-total transcendence of the family of all strongly minimal
theories in the expanded language $\Sigma'$, whereas adding to the
language $\Sigma$ infinitely many $0$-ary predicate symbols we
obtain the family $\mathcal{T}_\Sigma'$ with ${\rm
RS}(\mathcal{T}_\Sigma')=\infty$. Thus, Propositions \ref{prop1},
\ref{prop2}, \ref{prop3}, Corollary \ref{cor01}, and Facts
\ref{fact1}, \ref{fact2} imply the following:

\begin{theorem}\label{Thm} A family $\mathcal{T}_{\Sigma}$ of strongly minimal theories is $e$-totally transcendental if and only if the language $\Sigma$
consists of finitely many constant symbols as well as of finitely
many $0$-ary and unary predicate symbols.
\end{theorem}

\begin{note}
{\rm Assuming that distinct constant symbols are interpreted by
distinct elements, $e$-totally transcendental families of strongly
minimal theories in Theorem \ref{Thm} admit arbitrarily many
constant symbols in the language.}
\end{note}

\begin{note}
{\rm Since ${\rm RS}(\mathcal{T})\leq{\rm RS}(\mathcal{T}')$ for
$\mathcal{T}\subseteq\mathcal{T}'$ then the criterion for
$e$-total transcendence of families of strongly minimal theories
in Theorem \ref{Thm} can be trans\-formed for any family
$\mathcal{T}'$ containing the family $\mathcal{T}_{\Sigma}$ of
strongly minimal theories of given language $\Sigma$. In
particular, it holds for families of $\omega$-stable, superstable,
and stable theories.}
\end{note}


\bigskip

\end{document}
